\begin{table}%t2
\caption {Resolution and sensitivity for searches for radio SNRs
in nearby galaxies.\label{tab2}}
%\centerline
{\begin{tabular} {lllllll} \hline \hline\noalign{\smallskip}
Galaxy &  \hfil $\nu$    & Resolution & Resolution    & rms noise
&
Limiting $L_{\nu}$ & \\
&           (GHz)    & \hfil (arcsec) & \hfil (pc) &
($\mu$Jy/beam) &
($\times10^{22}{\rm erg\over s~ Hz}$) & Reference\\
\noalign{\smallskip} \hline LMC &1.40 & \hfil  912 &\hfil 243 &
\hfil 30 000& \hfil 11 & Filipovi\'{c} et~al. (\cite{Filipovic98}b)\\
SMC     &      1.42 & \hfil 828
&\hfil 261 &\hfil 15 000 & \hfil 7.7 & Filipovi\'{c} et~al. (\cite{Filipovic98}b)\\
IC 1613 &1.46 & \hfil  5 &\hfil 17  &  \hfil 56 &\hfil 3.2 &
Lozinskaya et~al. (\cite{Lozinskaya98})\\
M~31 & 1.465 & \hfil 5 &\hfil 18 &\hfil 30 & \hfil 2.0 &
Braun \& Walterbos (\cite{Braun93})\\
M~33 & 1.42 &   \hfil 7  &\hfil 28 &   \hfil 50&\hfil 4.1 & Gordon
et~al.
(\cite{Gordon99})\\
NGC 300 & 1.45 & \hfil 4 &\hfil 39 &\hfil 60 &\hfil 30 & Pannuti
et~al.
(\cite{Pannuti00})\\
NGC 6946 & 1.45 &  \hfil 2  &\hfil 49 &   \hfil 20 &\hfil 63
& Lacey et~al. (\cite{Lacey97})\\
NGC 7793 & 1.47 & \hfil  7 &\hfil 115&\hfil 60&\hfil 83 &
Pannuti et~al. (\cite{Pannuti02})\\
M~82& 8.4 & \hfil 0.182 &\hfil 3& \hfil  360&\hfil 660 &
Huang et~al. (\cite{Huang94})\\
NGC 1569 & 1.412 &\hfil 0.20 &\hfil 2 &\hfil 25 &\hfil 15 &
Greve et~al. (\cite{Greve02})\\
NGC 2146 & 1.6 &       \hfil 0.17  &\hfil 12  &\hfil 35 &\hfil 890
& Tarchi et~al. (\cite{Tarchi00})\\
 \hline
\end{tabular}}
\end{table}